

Citation: Sokolow SH, Wood CL, Jones IJ, Swartz SJ, Lopez M, Hsieh MH, et al. (2016) Global Assessment of Schistosomiasis Control Over the Past Century Shows Targeting the Snail Intermediate Host Works Best. PLoS Negl Trop Dis 10(7): e0004794. doi:10.1371/journal.pntd.0004794

Editor: Conor R. Caffrey, Center for Discovery and Innovation in Parasitic Diseases, UNITED STATES

Received: July 16, 2015

Accepted: May 31, 2016

Published: July 21, 2016

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative Commons CC0</u> public domain dedication.

Data Availability Statement: Data, including schistosomiasis-control fact sheets for the countries and territories assessed, are available at the purl website hosted on Stanford University library servers for public access: <u>https://purl.stanford.edu/</u> yt060bn1019.

Funding: All funding sources: a grant by the Woods Institute's Environmental Ventures Projects at Stanford University <u>https://woods.stanford.edu</u>, the Stanford Institute for Innovation in Developing Economies <u>https://seed.stanford.edu/</u> through the Global Development and Poverty Initiative **RESEARCH ARTICLE**

Global Assessment of Schistosomiasis Control Over the Past Century Shows Targeting the Snail Intermediate Host Works Best

Susanne H. Sokolow^{1,2}*, Chelsea L. Wood^{3,4}, Isabel J. Jones¹, Scott J. Swartz¹, Melina Lopez¹, Michael H. Hsieh^{5,6,7}, Kevin D. Lafferty^{2,8}, Armand M. Kuris², Chloe Rickards⁹, Giulio A. De Leo¹

1 Hopkins Marine Station, Stanford University, Pacific Grove, California, United States of America, 2 Marine Science Institute, and Department of Ecology, Evolution and Marine Biology, University of California, Santa Barbara, Santa Barbara, California, United States of America, 3 Michigan Society of Fellows, University of Michigan, Ann Arbor, Michigan, United States of America, 4 Department of Ecology and Evolutionary Biology, University of Michigan, Ann Arbor, Michigan, United States of America, 6 The George Washington University, Washington, D. C., United States of America, 6 The George Washington University, Washington, D. C., United States of America, 7 Biomedical Research Institute, Rockville, Maryland, United States of America, 8 Western Ecological Research Center, U.S. Geological Survey, Santa Barbara, California, United States of America, 9 Department of Bioengineering, Stanford University, Stanford, California, United States of America

* ssokolow@stanford.edu

Abstract

Background

Despite control efforts, human schistosomiasis remains prevalent throughout Africa, Asia, and South America. The global schistosomiasis burden has changed little since the new anthelmintic drug, praziquantel, promised widespread control.

Methodology

We evaluated large-scale schistosomiasis control attempts over the past century and across the globe by identifying factors that predict control program success: snail control (e.g., molluscicides or biological control), mass drug administrations (MDA) with praziquantel, or a combined strategy using both. For data, we compiled historical information on control tactics and their quantitative outcomes for all 83 countries and territories in which: (i) schistosomiasis was allegedly endemic during the 20th century, and (ii) schistosomiasis remains endemic, or (iii) schistosomiasis has been "eliminated," or is "no longer endemic," or transmission has been interrupted.

Principal Findings

Widespread snail control reduced prevalence by $92 \pm 5\%$ (N = 19) vs. $37 \pm 7\%$ (N = 29) for programs using little or no snail control. In addition, ecological, economic, and political factors contributed to schistosomiasis elimination. For instance, snail control was most

http://fsi.stanford.edu/docs/global-development-andpoverty-gdp, the Bill and Melinda Gates Foundation http://www.gatesfoundation.org, National Institutes of Health grant DK087895 and grant 1R01TW010286 http://www.nih.gov, and National Science Foundation Coupled Natural and Human Systems grant 1414102 http://www.nsf.gov. All grants provided partial salary support for SHS, IJJ, SJS, ML, MHH, AMK, and GADL. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared no competing interests exist.

common and widespread in wealthier countries and when control began earlier in the 20th century.

Conclusions/Significance

Snail control has been the most effective way to reduce schistosomiasis prevalence. Despite evidence that snail control leads to long-term disease reduction and elimination, most current schistosomiasis control efforts emphasize MDA using praziquantel over snail control. Combining drug-based control programs with affordable snail control seems the best strategy for eliminating schistosomiasis.

Author Summary

Schistosomiasis is a parasitic disease infecting more than 250 million people worldwide, with almost 800 million at risk. Over the past century, nations undertook schistosomiasis control programs, with outcomes varying from little effect to elimination. The biggest hope for elimination began about 40 years ago with the discovery of the antischistosomal drug praziquantel, after which snail control was seen as old fashioned. Here, we review control program outcomes over the past 100 years across all major schistosomiasis endemic zones, including Africa, Asia, and the Americas. We screened for differences in long-term schistosomiasis reductions among countries and found the most successful programs focused on transmission control (most often snail control, with or without engineering interventions), sometimes in tandem with praziquantel. Although praziquantel has important human-health benefits, our results suggest old-fashioned snail control has been the key to schistosomiasis elimination.

Introduction

Can we do better at controlling schistosomiasis? Despite effective drug treatment options and large-scale drug distribution programs, most endemic areas have not yet achieved satisfactory schistosomiasis control. Today, schistosomiasis remains prevalent in Africa, Asia, and South America where trends over time forecast perpetual endemicity. Sometimes, endemicity has been because poverty constrains control efforts; otherwise, endemicity is due to failed or ineffective control attempts. With more than 250 million people still infected and elimination stalled [1,2], the World Health Assembly (WHA) called for researching and applying complementary, non-pharmaceutical control strategies for eliminating schistosomiasis in its 2012 resolution 65.21 [3]. This resolution ignited debate over the best strategies for eliminating schistosomiasis [4–9]. To add quantitative data to this debate, we evaluated schistosomiasis control strategies over the past century and screened for factors associated with elimination or long-term prevalence reductions.

Human schistosomiasis occurs where aquatic (or amphibious) intermediate host snails shed infective *Schistosoma* spp. cercariae that penetrate human skin upon contact. Infected humans suffer from anemia, stunted growth, cognitive impairment, fatigue, infertility, and sometimes, liver fibrosis or bladder cancer [10]. Most affected people live in poverty where there are few resources for research and control [11].

Schistosomiasis control efforts aim to disrupt the parasite's complex life cycle ($\underline{Fig 1}$): sanitation stops parasite eggs in urine or feces from moving into aquatic snail habitats; snail control



Fig 1. Control strategies used against schistosomiasis during the past century.

reduces intermediate host density (parasite larvae reproduce asexually in snails); education (or "information education and communication"; IEC) helps people avoid high-risk water contacts and know when to seek treatment; and drugs—given as mass drug administrations (MDA), targeted treatment campaigns ("test-and-treat" or TAT), or through health services—kill the adult worm in the human host, with immediate and long-term health benefits for infected individuals [12]. Such efforts seem simple, but implementation often fails for economic or political reasons.

Consensus on the "best" schistosomiasis control strategy has varied over the past century. Early Egyptian control efforts around the turn of the 20th century treated human infection, then shifted toward borehole latrines as a sanitary measure in the 1930s, but with little success [13]. In the 1940s the focus shifted again, this time toward snail control using copper sulfate [13]. Control strategy recommendations by the World Health Organization (WHO) then emphasized integrated control measures, including clean water access, sanitation, snail control, health education, and health services, in addition to drug treatments [12]. In the late 1970s and early 1980s –after praziquantel became the drug-of-choice for treating schistosomiasis [14]– the integrated approach was often supplanted by chemotherapy via MDA. Praziquantel is safe and effective against adult schistosomes [15], but is ineffective against juvenile worms [16–18], and drug treatment does not prevent reinfection [19–23]. Nonetheless, large-scale research projects and influential modeling results suggested that widespread drug treatment (best when coupled with sanitation measures) would reduce schistosomiasis more than other interventions [24–27]. MDA increased in the 1990s after generic (inexpensive) praziquantel became available [15,28]. Thus, in 2001, the WHA endorsed preventive chemotherapy as the primary strategy to



Fig 2. Strategy details for schistosomiasis control: the current paradigm and an alternative based on past successes. (A) The current paradigm for global schistosomiasis control, adapted from [30], and (B) an alternative strategy based on historical successes (this paper). MDA = mass drug administration, TAT = targeted ("test-and-treat") chemotherapy.

control schistosomiasis through reducing morbidity associated with high worm burdens [29]. Specifically, Engels et al. [30] summarized the modern, WHO-recommended schistosomiasis global control strategy as MDA in high-transmission areas to reduce morbidity and transmission reduction in low-transmission areas (Fig 2a).

Although MDA is now the most popular control strategy, some have argued that snail control is more effective [5,6,31–33]. However, a Center for Global Development working group focused on successful global health interventions points out: "We don't know enough about what's worked because scaled up programs are rarely evaluated systematically" [34]. Here, using objective criteria and a quantitative analysis to test for commonalities among successful control programs, we find snail control has been effective at reducing schistosomiasis.

Methods

Defining Success

We evaluated control programs for all areas around the world with active (autochthonous) human schistosomiasis transmission at some time in the 20th century. We considered countries with little to no control effort as having *minimal control*. However if minimal control corresponded to a loss of schistosomiasis, we defined the outcome as *fortuitous elimination*. We defined control as *not (yet) successful* where (i) control has been incomplete, (ii) transmission continues, or (iii) the disease has been almost, but not yet, eliminated. We defined control as *successful* for active programs that were reported to have stopped local transmission (i.e., elimination or becoming "non-endemic") in WHO reports or peer-reviewed assessments (e.g., [9,35]). "Elimination" implies reducing disease incidence to zero in a particular area [30,36]. There is some inconsistency in the literature on the term "eradication," which often refers to global disease extirpation [37]. With respect to schistosomiasis, this has been applied to the regional elimination achieved by Japan. Here, because elimination, "eradication," and non-endemicity all imply no local transmission, we treated these designations as *successful*.

Countries and Territories Evaluated

Our goal was to evaluate control success in all countries and territories with endemic schistosomiasis during the 20th century. We began with the nine countries often cited as "success stories" for schistosomiasis elimination: Iran, Japan, Lebanon, Malaysia, Martinique, Montserrat, Thailand, Tunisia, and Turkey [<u>38</u>]. Antigua, Jordan, and Morocco were three other potential "successes" [<u>35,39</u>]. We were also interested in countries that achieved great reductions in schistosomiasis prevalence including Brazil, China, the Philippines, and Egypt. Additional literature searches focused on characterizing disease and control history for all additional countries with: (i) historical disease data and (ii) recorded national- or territory-level schistosomiasis control programs. Although we found relevant data for most countries, data were contradictory for several Caribbean islands such as Guadeloupe and Dominican Republic, with some reports indicating elimination and others claiming ongoing risk. We considered these countries to be "not (yet) successful." For several in-conflict countries such as Chad and Syria, current schistosomiasis prevalence is "unknown", with the potential for conflict and political unrest to hinder control [40].

Data Collection

We obtained country-specific data for several categories (<u>S1 Table</u>) by reading peer-reviewed published sources as well as non-peer-reviewed reports accessed through online (or hard copy) repositories, including PubMed, ISI Web of Science, Google Scholar, WHO, United Nations (UN), World Bank, United States Agency for International Development, and the UN Food and Agriculture Organization (see <u>S1 Appendix</u> for complete reference list). From these sources, we assessed 77 countries and six semi-autonomous territories (including Western Sahara in north-ern Africa, Guadeloupe, Martinique, Montserrat, Puerto Rico, and Zanzibar).

For each country, we collected information on schistosomiasis, control efforts, parasite life cycles, environmental factors, and economics. We focused on variables related to national schistosomiasis data (country- or territory-wide prevalence, infected population size, at-risk population size) and details about the control strategies implemented and their time-course. We also recorded snail and schistosome species present; island or mainland geography; and per-capita gross domestic product (GDP) in 2013 and in all years for which schistosomiasis disease data were available in each country. Further, we noted site-specific factors that might alter disease outcomes or resources for control activities (S1 Table). We limited the prevalence information, in almost every case, to country-level (or territory-level) statistics. Only for Japan, where finer scale data were available over many years, did we use large-scale and long-term regional data to assess trends, and we included only the data from the largest endemic area (the Kofu basin) in the statistical analyses. We were careful to avoid small-scale, focal studies on prevalence that might not represent the whole country. We treated countries/territories as replicates in statistical analyses done in JMP Pro version 12 [41] and R version 3.1.2 [42].

Statistics

To test the general hypothesis that control programs can eliminate schistosomiasis, we assessed whether schistosomiasis was eliminated/non-endemic using a logistic regression, with five predictors for 68 countries/territories (excluding 15 that lacked enough data or were designated non-endemic to begin with): (i) the presence/absence of a national- or territory-level control program; (ii) status as a mainland or island (because it should be easier to achieve elimination with more isolation); (iii) the total human population infected with schistosomiasis when the starting infected population is large); (iv) the current fraction of people with access to improved water sources (as a proxy for contemporary water, sanitation, and hygiene conditions, World Bank Development Indicators, 2012 [43]); and (v) contemporary per-capita GDP (as a proxy for "wealth" status, World Bank Development Indicators, 2013 [44]; S2 Table). Using 68 countries gave us considerable statistical power to evaluate when and where control has been effective in eliminating schistosomiasis.

We next compared how well different control strategies reduced disease. The strategies used in historical schistosomiasis control efforts were categorized as: MDA, snail control, or engineering interventions (e.g., sanitation infrastructure, cement lined canals, drained wetlands). Each control category was further sorted according to our best estimate (based on qualitative descriptions, or sometimes, quantitative reports): extensive/complete (>70% of the population/ area in need received treatment), intermediate (>30%), or focal to none (<30%). Although disease can be measured as intensity [45,46] (as indicated by patient egg output), there was not enough published national-level data to assess intensity means and variances. Therefore, we compared disease prevalence on a continuous scale (0 to 100% based on the schistosomiasis national prevalence at each available time point for each country/territory). We included only countries with national control programs and enough longitudinal disease data. We excluded those countries/territories with no coordinated control effort ("Minimal control" in Table 1, Fig 3). Further, among national control programs, we designated the category "low coverage" where snail control, or MDA, or both were included, but the program achieved low (<30%) coverage for either strategy. Similarly, engineering controls were considered present only where their coverage was high (i.e. where more than 30% coverage was achieved).

To test the hypothesis that control strategies differed in their ability to reduce prevalence, we used a quantitative generalized linear mixed model (GLMM, function "glmer" from the R package "lme4" ([42], S2 Table). The statistical model assessed what factors best predicted relative change in prevalence over time for the 44 countries that applied concerted control and had quantitative, longitudinal data on prevalence, control strategies and covariates (more details below and in S2 Table). This GLMM considered country as a random effect (to account for the repeated measures over time within each country) and the following fixed effects: (i) control program duration (to test if longer efforts might be more successful); (ii) a country's status as an island or mainland; (iii) initial prevalence before control began (to account for the control effort needed); (iv) the percentage of the population with access to improved water sources (World Bank Indicators, 2012 [43]); and (v) the inflation-adjusted per capita GDP over time (recorded at each time point with disease data from The Maddison-Project [47]). We were most interested in the interaction terms between the predictors and time [year], which, if significant, would indicate an effect on prevalence reduction or increase over time. We first assembled a "full model" that contained all predictors and interaction terms, and then used a model selection procedure based on Akaike's information criterion (AIC) to remove each interaction term in turn to find the best balance between parsimony and fit to the data ([48], S3Table).

After analyzing what control strategies were most successful, we became curious about the factors that might have determined which control strategies a country used. To that end, we assessed the correlations between the control strategies used and a country's "wealth" status (per-capita GDP for each country at each time-point) as well as the control era (the year each national- or territory-level control program began).

Results

The response to schistosomiasis varied across the 83 evaluated countries/territories. Seventytwo countries and five territories were "endemic" for schistosomiasis during the 20th century (whereas four were probably "never endemic," and two had too little data to determine their historical or contemporary schistosomiasis endemicity; <u>Table 1</u>, Fig 3). Only 51 (66%) of endemic countries/territories had coordinated national- or territory-level schistosomiasis control during the past century, whereas the remaining 26 (34%) had no verifiable programs (despite some having small pilot programs; <u>Table 1</u>, Figs <u>3</u> and <u>4</u>). Nine endemic countries/

Country/ territory name	Historical or contemporary national schistosomiasis control program?	Control program outcome (success category)	Prevalence reduc	Prevalence reduction (%)		Population at risk reduction (%)	
			Baseline: post- control	Baseline: now	Baseline: post- control	Baseline: now	
Algeria	ND	Not enough data	ND	ND	ND	ND	
Angola	No	Minimal control	NA	ND	NA	ND	
Antigua	No	Fortuitous elimination	NA	100	NA	100	
Benin	Yes	Not (yet) successful	-2	-2	ND	ND	
Botswana	No	Minimal control	NA	NA	NA	NA	
Brazil	Yes	Not (yet) successful	92	80	72	69	
Burkina Faso	Yes	Not (yet) successful	61.2	61.2	ND	ND	
Burundi	Yes	Not (yet) successful	74.4	74.4	ND	ND	
Cambodia	Yes	Not (yet) successful	83	83	90	90	
Cameroon	Yes	Not (yet) successful	16.7	16.7	ND	ND	
Cape Verde		Never endemic	NA	NA	NA	NA	
Cen. African Republic	Yes	Not (yet) successful	-58	-58	ND	ND	
Chad	No	Minimal control	NA	NA	NA	ND	
China	Yes	Not (yet) successful	98.9	98.9	79	79	
Comoros		Never endemic	NA	NA	NA	NA	
Congo	Yes	Not (yet) successful	-58	41.7	ND	ND	
Côte d'Ivoire	No	Minimal control	NA	NA	NA	NA	
Dem. Rep. of the Congo	No	Minimal control	NA	NA	NA	NA	
Djibouti	No	Fortuitous elimination	NA	NA	NA	NA	
Dominican Republic	Yes	Not (yet) successful	ND	ND	ND	ND	
Egypt	Yes	Not (yet) successful	99	99	ND	ND	
Equatorial Guinea	No	Minimal control	NA	NA	NA	NA	
Eritrea	No	Minimal control	NA	NA	NA	NA	
Ethiopia	No*	Minimal control	NA	NA	NA	NA	
Gabon	No	Minimal control	NA	NA	NA	NA	
Gambia	No	Minimal control	NA	NA	NA	NA	
Ghana	Yes	Not (yet) successful	73.9	73.9	ND	ND	
Guadeloupe	Yes	Not (yet) successful	38	96	-16	ND	
Guinea	No*	Minimal control	NA	NA	NA	NA	
Guinea-Bissau	No	Minimal control	NA	NA	NA	NA	
Indonesia	Yes	Not (yet) successful	99.5	99.5	90	ND	
Iran	Yes	Successful	99.5	100	88.6	100	
Iraq	Yes	Not (yet) successful	99.4	99.5	63	ND	
Japan	Yes	Successful	100	100	100	100	
Jordan	Yes	Successful	100	100	100	100	
Kenya	Yes	Not (yet) successful	51	51	ND	ND	
Laos	Yes	Not (yet) successful	96	84.6	4.7	ND	
Lebanon	Yes	Successful	100	100	100	100	
Lesotho	No	Minimal control	NA	NA	NA	NA	
Liberia	No*	Minimal control	ND	ND	ND	ND	

Table 1. Countries and territories evaluated and their success categories, prevalence reduction, and percent reduction in population at risk for schistosomiasis (as a proportion of the total population).

(Continued)

Table 1. (Continued)

Country/ territory name	Historical or contemporary national schistosomiasis control program?		Prevalence reduc	Prevalence reduction (%)		Population at risk reduction (%)	
		Control program outcome (success category)	Baseline: post- control	Baseline: now	Baseline: post- control	Baseline now	
Libya	Yes	Not (yet) successful	66.7	66.7	NA	NA	
Madagascar	Yes	Not (yet) successful	5.4	73.8	-14.6	ND	
Malawi	Yes	Not (yet) successful	81.8	43.3	ND	ND	
Malaysia	No	Fortuitous elimination	NA	NA	NA	NA	
Mali	Yes	Not (yet) successful	50.6	51.8	-88	ND	
Martinique	Yes	Successful	88.9	100	100	100	
Mauritania	No	Minimal control	NA	NA	NA	NA	
Mauritius	Yes	Successful	35.7	100	0	100	
Montserrat	No	Fortuitous elimination	NA	NA	NA	NA	
Morocco	Yes	Successful	100	100	100	100	
Mozambique	Yes	Not (yet) successful	28.9	28.9	ND	ND	
Namibia	No*	Minimal control	NA	NA	NA	NA	
Niger	Yes	Not (yet) successful	50	44	ND	ND	
Nigeria	No	Minimal control	NA	NA	NA	NA	
Oman	Yes	Not (yet) successful	3	0.6	-1025	ND	
Pakistan		Never endemic	NA	NA	NA	NA	
Philippines	Yes	Not (yet) successful	93	98.3	11	ND	
Puerto Rico	Yes	Successful	85.3	100	78.4	100	
Rwanda	Yes	Not (yet) successful	69.5	69.5	ND	ND	
Sao Tome & Principe	No	Minimal control	NA	NA	NA	NA	
Saudi Arabia	Yes	Not (yet) successful	81.05	99.8	-18	ND	
Senegal	Yes	Not (yet) successful	1	1	ND	ND	
Seychelles		Never endemic	NA	NA	NA	NA	
Sierra Leone	Yes	Not (yet) successful	51.4	51.4	ND	ND	
Somalia	Yes	Not (yet) successful	-24	-24	NA	NA	
South Africa	No	Minimal control	NA	NA	NA	NA	
St. Lucia	Yes	Not (yet) successful	88	98.2	84.3	84.3	
Sudan	Yes	Not (yet) successful	-29.7	-29.7	47	47	
Surinam	Yes	Not (yet) successful	61.5	61.5	69.3	69.3	
Swaziland	Yes	Not (yet) successful	9.6	9.6	ND	ND	
Syria	Yes	Not (yet) successful	65.4	ND	38	ND	
Tanzania	Yes	Not (yet) successful	60	60	0	ND	
Thailand	No	Fortuitous elimination	NA	NA	NA	NA	
Тодо	Yes	Not (yet) successful	30.9	30.9	ND	ND	
Tunisia	Yes	Successful	100	100	100	100	
Turkey	No	Fortuitous elimination	NA	NA	NA	NA	
Uganda	Yes	Not (vet) successful	55.4	55.4	ND	ND	
Venezuela	Yes	Not (yet) successful	90	98.6	ND	ND	
Western Sahara	ND	not enough data	ND	90.0 ND	ND	ND	
Yemen	Yes	Not (yet) successful	44	44	ND	ND	
Zambia	Yes	Not (yet) successful	26.6	26.6	ND	NE	
Zanzibar	Yes	Not (yet) successful	76.6	84.7	ND	NE	

(Continued)

Table 1. (Continued)

			Prevalence reduction (%)		Population at risk reduction (%)		
Country/ territory name	Historical or contemporary national schistosomiasis control program?	Control program outcome (success category)	Baseline: post- control	Baseline: now	Baseline: post- control	Baseline: now	
Zimbabwe	No*	Minimal control	NA	NA	NA	NA	

Negative values represent increases.

"Baseline: post-control" compares just before to just after the control program.

"Baseline: now" refers to just before compared with contemporary estimates.

NA = not applicable.

ND = no data.

*Indicates that a control program has begun, but too recently (2012 or later) to evaluate its nationwide effect: in Ethiopia, control began 2015; in Guinea, control began 2012; in Liberia, control began 2012; in Namibia, control using praziquantel is planned but not yet started in 2015; in Zimbabwe, control began 2012)

doi:10.1371/journal.pntd.0004794.t001



Fig 3. Countries and territories evaluated and their outcomes for schistosomiasis control or elimination. "Successful" = there was a control program that preceded elimination or non-endemic status; "Fortuitous elimination" = elimination or non-endemic status with no control program; "Minimal control" = endemic disease in the face of minimal to no control, even if there were some pilot or small-scale programs; "Not (yet) successful" = endemic disease with a past or present control program.

doi:10.1371/journal.pntd.0004794.g003





territories—Iran, Japan, Jordan, Lebanon, Martinique, Mauritius, Morocco, Puerto Rico, and Tunisia—applied coordinated control programs and achieved "success" as we defined it here.

Nine countries/territories that have not yet achieved schistosomiasis elimination achieved a 90% or greater reduction in their country-level prevalence since baseline (before control): China, Egypt, Guadeloupe, Indonesia, Iraq, Philippines, Saudi Arabia, St. Lucia, and Venezuela. The remaining countries had variable schistosomiasis prevalence over the past century (Table 1), depending, in part, on their control strategy.

Predictor	Estimate	p value
Control program?-presence/absence	1.22	0.42
Population infected before control (log-transformed)	-0.74	0.028*
Improved water source (rural % with access), 2012	0.16	0.027*
GDP per capita (2013)	-0.00004	0.55
Island? Yes/no	-0.15	0.87

Table 2. Logistic regression for elimination/non-endemicity. *

*see S2 Table for more statistical details; N = 68 countries/territories evaluated

doi:10.1371/journal.pntd.0004794.t002

Elimination Predictors

Counter to expectations, elimination/non-endemicity was not associated with having a control program. This unexpected result was due to two factors: several countries/territories achieved "fortuitous" elimination without any documented control effort (Antigua, Djibouti, Malaysia, Montserrat, Thailand, and Turkey) and several other countries failed to eliminate schistosomiasis, despite substantial prevalence reductions. Island/mainland did not predict elimination status (Table 2), however, our inclusion of population size, which is higher on continents and makes elimination harder, could have co-varied with a mainland-island effect. Possessing greater "wealth" (indicated by a higher contemporary per capita GDP) did not affect elimination. Elimination was, however, more likely where more people can access improved water sources. In summary, achieving elimination was idiosyncratic. It was easier with smaller infected populations and in countries with improved (safer) water sources. Although many programs have failed to eliminate schistosomiasis, sometimes elimination has occurred without a coordinated control program. Below, we discuss what factors in addition to control programs could affect schistosomiasis prevalence reductions and elimination success.

Effective Control Strategies—Quantitative Outcomes

Although fortuitous elimination in several countries confounded our ability to assess whether control programs eliminated schistosomiasis, many areas with control programs experienced durable *prevalence reductions*. A program's effectiveness (i.e., the prevalence reduction rate) depended strongly on strategy type and coverage and weakly on the intercept (prevalence at baseline). Applying snail control, MDA, or both—with at least intermediate (>30%) coverage —worked better than any programs with low coverage. Snail control programs (primarily mollusciciding and biological control using non-native, competitor snails) showed the strongest prevalence reductions (while accounting for other covariates, including: control duration [in years], country "wealth" [as per capita GDP in each year with disease data], and access to improved water sources; <u>Table 3</u>, Figs <u>5</u> and <u>6</u>). In other words, all else being equal, prevalence reduction was highest with snail control at intermediate or better coverage.

Although engineering controls (e.g., installing sanitation infrastructure, cementing canals, building bridges, or draining wetlands), were almost always accompanied by snail control, about half of the programs using snail control did not use engineering in their control programs. Programs that used MDA as a primary strategy (without snail control) also did not report using any large-scale engineering controls. The presence or absence and extent of engineering controls showed weak effects on prevalence, and including 3-way interactions with this variable—along with the other control strategies and time—in the quantitative statistical model did not improve model fit to the data (based on AIC; <u>S3 Table</u>). Thus, engineering controls,

Table 3. Generalized linear mixed model (GLMM, see <u>S2</u> and <u>S3</u> Tables) comparing change in prevalence for control programs using: MDA with
praziquantel, snail control primarily (snail control), both (MDA + snail control), or minimal control (<30% coverage, "low coverage," not shown).

Description	Predictor	Estimate* (X 10 ⁻²)	p-value
Interaction terms**	Control duration:time	0.02	<0.001
	Island:time	1.1	<0.001
	Improved water (contemporary % with access):time	0.09	<0.001
	GDP per capita (inflation adjusted):time	1.2	0.0013
	Baseline prevalence:time	-0.04	<0.001
	Engineering controls:time	0.09	0.79
	MDA:time	-2.3	<0.001
	Snail control:time	-12.3	<0.001
	Both (MDA + snail control):time	-7.9	<0.001

*The estimate indicates the change in prevalence over time where more negative (positive) values indicate stronger prevalence reduction (increase) associated with that predictor.

Random effects for country/territory (intercept): variance = 2.06, std. dev. 1.44

N = 44 countries/territories evaluated.

**These are predictors association with change in prevalence over time (years) in each country/territory

doi:10.1371/journal.pntd.0004794.t003

although perhaps beneficial within some integrated programs, did not consistently reduce schistosomiasis prevalence.

Population size affected control success. As expected, prevalence reductions were impaired where there were larger initial infected human populations, but this relationship differed among the control strategies. Snail control programs (with or without MDA) were less sensitive to initial infected human population size, than were other approaches (Fig 7).

Wealth and Era Effects

Control strategy depended on country wealth and the year in which control began. Richer countries (measured by inflation-adjusted, per-capita GDP) tended to begin their control programs earlier in the 20th century, with a stronger focus on snail control and greater success (Fig <u>8</u>). Higher wealth was also correlated with greater access to improved (safer) water sources.

Large-scale MDA programs were rare before praziquantel entered the global market in the 1980s. After this turning point, there was a new option (using both MDA and snail control) and this integrated strategy has been used since the 1980s in places like China, Egypt, and Brazil. Countries that began their control programs even more recently (after the 1990s or 2000s) were poorer and tended either to focus on MDA or achieved poor coverage (designated as "low coverage" in Fig 8).

Discussion

Our results support recent suggestions that snail control is key to schistosomiasis reduction [5,6,49]. Such an effect has been anticipated. In 1985, a lead researcher of the Caribbean "St. Lucia Project,"-a Rockefeller-funded schistosomiasis control study—wrote, "chemother-apy is now assuming the major role in control programmes, but in most... a reservoir of infection inevitably remains. Transmission is thus likely to continue at a low, but probably increasing level unless a supplementary control strategy is present" [24]. Unfortunately, it took decades to assess this prediction.

Chemotherapy has major benefits for infected humans, but, by itself, MDA has done little to curb re-infection. Although programs limited to MDA with praziquantel did not appear to



Fig 5. Schistosomiasis prevalence change over time. Prevalence change by control program strategy (time 0 on the x-axis is set when control began; negative values for the normalized year show data *n* years before control started and positive values *n* years after).







doi:10.1371/journal.pntd.0004794.g007



mean inflation-adjusted, per-capita GDP throughout the relevant time for which disease data were available, and bars represent the range.

do as well as the other strategies evaluated, the (targeted) chemotherapy for infection control remains an undeniable factor in improving health, especially when integrated with snail control. Countries whose programs focused on snail control often relied on distributing chemotherapy through means other than MDA, such as Morocco's successful test-and-treat (TAT) campaigns using mobile teams [50], Iraq's early school-based TAT programs [51], and Japan's involvement as an early TAT site for praziquantel beginning in the late 1970s. This involvement might have carried Japan to country-wide elimination by 1996 [52]. One reason praziquantel seems less effective than expected is that it was applied later in history when control campaigns targeted more challenging countries. In other words, schistosomiasis elimination was more successful among programs started before praziquantel reached the global market than among those programs started after the drug's introduction in the late 1970s. This might arise, in part, because wealthier countries tended to address the disease earlier in the 20th century, as they could afford molluscicides for widespread snail control (although inexpensive biological control also sometimes succeeded).

The "fortuitous elimination" of schistosomiasis from Antigua, Djibouti, Malaysia, Montserrat, Thailand, and Turkey without documented control efforts suggests cryptic factors have affected schistosomiasis, including: species invasions (e.g. snail competitors or predators), sanitation or health care improvements outside control programs, and human-induced or natural ecosystem changes (such as changes in dams, irrigated-agriculture, and urbanization). The least fortuitous of the fortuitous eliminations was the 1995 volcanic eruption that drove almost half of Montserrat's population off the island and made the schistosomiasis transmission zones off limits to people [53,54]. (See <u>S2 Appendix</u> and <u>S1 Fig</u> for more cryptic schistosomiasis control examples).

These results suggest that programs have been most effective when snail control is coordinated soon thereafter—or simultaneously—with chemotherapy (morbidity control) via a rational progression from widespread, active drug distribution campaigns (MDA or targeted treatment (TAT)) to a focus on high-risk groups and finally passive distribution within health services coupled with surveillance (e.g. "surveillance and response"[55,56]) and health education (e.g. IEC) in the "end game" (Fig 2b).

As for how to control snails, the most common strategy has been to use expensive and toxic molluscicides; an effort that is neither feasible nor desirable for many poor countries where schistosomiasis is now endemic. Schistosomiasis has been hard to control without well-funded, national-level efforts, and the contemporary global health discussion has been focused on strategies that optimize efficiency and affordability. By recognizing the successful use of snail control for transmission reduction, and by fostering research directed toward the development of creative, safe and cheap tools to target the snail intermediate host, global schistosomiasis elimination might be attainable.

Supporting Information

S1 Appendix. Additional references. (DOCX)

S2 Appendix. Supporting text. Detailed examples of cryptic social, ecological, and political factors: species invasions, sanitation, and ecosystem change. (DOCX)

S1 Table. List of variables searched for and recorded (when available) to assemble the database analyzed in this paper. (DOCX)

S2 Table. Variables included (and regression formulas used) in the binary and quantitative analyses.

(DOCX)

S3 Table. Model selection results leading to the final model discussed in the main text. "3way interactions" signify a Strategy*Year*Engineering term in the full-model. The other variables are as in <u>S2 Table</u>. Those models above the dotted line were deemed to fit the data best. Removing the 3way interactions with engineering controls did not substantially change the model fit to data, whereas removing the other variables decreased the fit (increased Akaike's information criterion (AIC)) substantially. (DOCX)

S1 Fig. Progression of schistosomiasis prevalence reductions (black lines) and timing of alien species introductions (intentional or inadvertent) which led to biological invasions of competitor snails (solid arrows) in the Caribbean or crayfish (dashed arrow) in Egypt. The suspected time-course of biological invasions in these regions—with most invasions occurring just before or during periods of maximal prevalence reduction—suggests the plausibility that invasions may have influenced schistosomiasis control outcomes for many, if not all, of these case studies.

(DOCX)

Acknowledgments

The authors thank Gregory T. Galin and Lee Marom for assistance with figures and artwork, Olivia Cords, Cassandra Coulter, Anika Knight, Marcus Munoz, Austin Rife, Morgan Rondinelli, and Alexis Wait, for assistance in data collection, Diana La for support with mapping and geographical information systems, Trace Hancock, Cheryl Butner, and Chris Patton for assistance in building websites for outreach and data repositories for this work, and Merlijn Jocque plus three anonymous reviewers for input on earlier manuscript drafts. Any use of trade, product, or firm names in this publication is for descriptive purposes only and does not imply endorsement by the U.S. government.

Author Contributions

Conceived and designed the experiments: SHS GADL CLW MHH. Performed the experiments: SHS IJJ SJS ML. Analyzed the data: SHS GADL CLW KDL CR. Wrote the paper: SHS GADL CLW SJS AMK KDL ML IJJ MHH.

References

- Lai YS, Biedermann P, Ekpo UF, Garba A, Mathieu E, et al. (2015) Spatial distribution of schistosomiasis and treatment needs in sub-Saharan Africa: a systematic review and geostatistical analysis. Lancet Infect Dis 15: 927–940. doi: <u>10.1016/S1473-3099(15)00066-3</u> PMID: <u>26004859</u>
- Hotez PJ, Alvarado M, Basanez MG, Bolliger I, Bourne R, et al. (2014) The Global Burden of Disease study 2010: interpretation and implications for the neglected tropical diseases. PLoS Negl Trop Dis 8: e2865. doi: <u>10.1371/journal.pntd.0002865</u> PMID: <u>25058013</u>
- 3. WHA (2012) Elimination of schistosomiasis Geneva: WHO: WHA65/2012/REC/1 Sixty-fifth World Health Assembly: resolutions and decisions annexes.
- Bockarie MJ, Kelly-Hope LA, Rebollo M, Molyneux DH (2013) Preventive chemotherapy as a strategy for elimination of neglected tropical parasitic diseases: endgame challenges. Philos Trans R Soc Lond B Biol Sci 368: 20120144. doi: 10.1098/rstb.2012.0144 PMID: 23798692
- Gray DJ, McManus DP, Li Y, Williams GM, Bergquist R, et al. (2010) Schistosomiasis elimination: lessons from the past guide the future. Lancet Infect Dis 10: 733–736. doi: <u>10.1016/S1473-3099(10)</u> <u>70099-2</u> PMID: <u>20705513</u>
- King CH, Bertsch D (2015) Historical perspective: snail control to prevent schistosomiasis. PLoS Negl Trop Dis 9: e0003657. doi: 10.1371/journal.pntd.0003657 PMID: 25905621
- Ross AG, Olveda RM, Li Y (2015) An audacious goal: the elimination of schistosomiasis in our lifetime through mass drug administration. Lancet 385: 2220–2221. doi: <u>10.1016/S0140-6736(14)61417-3</u> PMID: <u>25467574</u>
- Savioli L, Fenwick A, Rollinson D, Albonico M, Ame SM (2015) An achievable goal: control and elimination of schistosomiasis. Lancet 386: 739.
- Rollinson D, Knopp S, Levitz S, Stothard JR, Tchuem Tchuenté LA, et al. (2013) Time to set the agenda for schistosomiasis elimination. Acta Tropica 128: 423–440. doi: <u>10.1016/j.actatropica.2012.04.013</u> PMID: <u>22580511</u>
- King C, Dickman K, Tisch D (2005) Reassessment of the cost of chronic helmintic infection: a metaanalysis of disability-related outcomes in endemic schistosomiasis. Lancet 365: 1561–1569. PMID: 15866310
- 11. King CH (2010) Parasites and poverty: the case of schistosomiasis. Acta Trop 113: 95–104. doi: <u>10.</u> <u>1016/j.actatropica.2009.11.012</u> PMID: <u>19962954</u>
- WHO (1993) The control of schistosomiasis. Second report of the WHO expert committee. WHO Tech Rep Ser 830: 1–86.
- Sandbach FR (1976) The history of schistosomiasis research and policy for its control. Med Hist 20: 259–275. PMID: <u>792584</u>
- 14. WHO (1985) The control of schistosomiasis. Report of a WHO expert committee. WHO Tech Rep Ser 728: 1–113.
- Doenhoff M, Kimani G, Cioli D (2000) Praziquantel and the control of schistosomiasis. Parasitol Today 16: 364–366. PMID: <u>10951592</u>
- Gonnert R, Andrews P (1977) Praziquantel, a new board-spectrum antischistosomal agent. Z Parasitenkd 52: 129–150. PMID: <u>410178</u>
- Sabah AA, Fletcher C, Webbe G, Doenhoff MJ (1986) Schistosoma mansoni: chemotherapy of infections of different ages. Exp Parasitol 61: 294–303. PMID: <u>3086114</u>
- Pica-Mattoccia L, Cioli D (2004) Sex- and stage-related sensitivity of Schistosoma mansoni to in vivo and in vitro praziquantel treatment. Int J Parasitol 34: 527–533. PMID: <u>15013742</u>

- Chandiwana SK, Woolhouse ME, Bradley M (1991) Factors affecting the intensity of reinfection with Schistosoma haematobium following treatment with praziquantel. Parasitology 102 Pt 1: 73–83. PMID: 1903882
- Moloney NA, Hinchcliffe P, Webbe G (1987) Loss of resistance to reinfection with Schistosoma japonicum in mice after treatment with praziquantel. Trans R Soc Trop Med Hyg 81: 247–254. PMID: <u>3113003</u>
- N'Goran EK, Utzinger J, N'Guessan AN, Muller I, Zamble K, et al. (2001) Reinfection with Schistosoma haematobium following school-based chemotherapy with praziquantel in four highly endemic villages in Cote d'Ivoire. Trop Med Int Health 6: 817–825. PMID: <u>11679130</u>
- 22. Garba A, Lamine MS, Barkire N, Djibo A, Sofo B, et al. (2013) Efficacy and safety of two closely spaced doses of praziquantel against *Schistosoma haematobium* and *S. mansoni* and re-infection patterns in school-aged children in Niger. Acta Trop 128: 334–344. doi: <u>10.1016/j.actatropica.2012.08.008</u> PMID: <u>22940014</u>
- Webster BL, Diaw OT, Seye MM, Faye DS, Stothard JR, et al. (2013) Praziquantel treatment of school children from single and mixed infection foci of intestinal and urogenital schistosomiasis along the Senegal River Basin: monitoring treatment success and re-infection patterns. Acta Trop 128: 292–302. doi: 10.1016/j.actatropica.2012.09.010 PMID: 23022016
- Jordan P (1985) Schistosomiasis: the St. Lucia project. Cambridge, UK: Cambridge University Press. 451 p.
- Jordan P (1977) Schistosomiasis—research to control. Am J Trop Med Hyg 26: 877–886. PMID: 562074
- MacDonald G (1965) Dynamics of helminth infections with special reference to schistosomes. Trans Royal Soc of Trop Med Hyg 59: 489–506.
- Anderson RM, May RM (1979) Prevalence of schistosome infections within molluscan populations observed patterns and theoretical predictions. Parasitology 79: 63–94. PMID: <u>542322</u>
- Hagan P, Appleton CC, Coles GC, Kusel JR, Tchuem-Tchuente LA (2004) Schistosomiasis control: keep taking the tablets. Trends Parasitol 20: 92–97. PMID: <u>14747023</u>
- WHA (2001) Schistosomiasis and soil-transmitted helminth infections. Geneva: WHO: WHA54/2001 Fifty-fourth World Health Assembly.
- Engels D, Chitsulo L, Montresor A, Savioli L (2002) The global epidemiological situation of schistosomiasis and new approaches to control and research. Acta Trop 82: 139–146. PMID: <u>12020886</u>
- Sokolow SH, Lafferty KD, Kuris AM (2014) Regulation of laboratory populations of snails (*Biomphalaria* and *Bulinus* spp.) by river prawns, *Macrobrachium* spp. (Decapoda, Palaemonidae): implications for control of schistosomiasis. Acta Trop 132C: 64–74.
- 32. Knopp S, Person B, Ame SM, Mohammed KA, Ali SM, et al. (2013) Elimination of schistosomiasis transmission in Zanzibar: baseline findings before the onset of a randomized intervention trial. PLoS Negl Trop Dis 7: e2474. doi: 10.1371/journal.pntd.0002474 PMID: 24147165
- Sokolow SH, Huttinger E, Jouanard N, Hsieh MH, Lafferty KD, et al. (2015) Reduced transmission of human schistosomiasis after restoration of a native river prawn that preys on the snail intermediate host. Proc Natl Acad Sci U S A 112: 9650–9655. doi: 10.1073/pnas.1502651112 PMID: 26195752
- Alleyne G, Barrett S, Claeson M, Chowdhury M, Easterly W, et al. (2005) What's worked? Accounting for success in global health. Geneva, Switzerland: World Health Organization.
- 35. WHO (2011) Elimination of schistosomiasis: Report by the secretariat. Geneva: WHO.
- **36.** Dowdle WR (1998) The principles of disease elimination and eradication. Bull World Health Organ 76 Suppl 2: 22–25. PMID: <u>10063669</u>
- Molyneux DH, Hopkins DR, Zagaria N (2004) Disease eradication, elimination and control: the need for accurate and consistent usage. Trends Parasitol 20: 347–351. PMID: 15246314
- Steinmann P, Keiser J, Bos R, Tanner M, Utzinger J (2006) Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. Lancet Infect Dis 6: 411–425. PMID: 16790382
- EMRO/WHO (2007) Inter-country meeting on strategies to eliminate schistosomiasis from the Eastern Mediterranean Region. Muscat: Eastern Mediterranean Region Organization of the World Health Organization.
- 40. Bruun B, Aagaard-Hansen J (2008) The social context of schistosomiasis and its control: an introduction and annotated bibliography. Geneva, Switzerland: World Health Organization.
- 41. SAS (2013) JMP Proversion 11. Cary, NC: SAS Institute Inc.
- CRAN (2014) R Core Team: A language and environment for statistical computing. Vienna: R Foundation for Statistical Computing.

- Anon. World Bank Indicators: Improved water source, rural (% of population with access). WHO/UNI-CEF Joint Monitoring Programme for Water Supply and Sanitation (<u>http://wwwssinfoorg/</u>) 2013.
- 44. Anon. World Bank Indicators: GDP per capita (current US\$). 2013 [cited 2015 Dec 31]; Available from: http://data.worldbank.org/indicator/NY.GDP.PCAP.CD.
- 45. Montresor A (2011) Cure rate is not a valid indicator for assessing drug efficacy and impact of preventive chemotherapy interventions against schistosomiasis and soil-transmitted helminthiasis. Trans R Soc Trop Med Hyg 105: 361–363. doi: 10.1016/j.trstmh.2011.04.003 PMID: 21612808
- WHO (2013) Assessing the efficacy of anthelminthic drugs against schistosomiasis and soil-transmitted helminthiases. Geneva: WHO.
- The Maddison-Project. 2013 [cited 2015 Dec 31]; Available from: <u>http://www.ggdc.net/maddison/maddison-project/home.htm</u>.
- **48.** Burnham K, Anderson D (2002) Model selection and multimodel inference: a practical information-theoretic approach. New York: Springer-Verlag. 488 p.
- Utzinger J, Zhou XN, Chen MG, Bergquist R (2005) Conquering schistosomiasis in China: the long march. Acta Trop 96: 69–96. PMID: <u>16312039</u>
- Barkia H, Barkia A, Yacoubi R, Alemad A, El Kharim K, et al. (2014) Contribution of mobile teams to efforts to eliminate schistosomiasis at Schistosoma haematobium in Morocco—narrative review article. Iranian Journal of Public Health 43: 1167–1175. PMID: <u>26175970</u>
- Yacoub A, Southgate BA (1987) The epidemiology of schistosomiasis in the later stages of a control programme based on chemotherapy: the Basrah study. 1. Descriptive epidemiology and parasitological results. Trans R Soc Trop Med Hyg 81: 449–459. PMID: <u>3120370</u>
- Tanaka H, Tsuji M (1997) From discovery to eradication of schistosomiasis in Japan: 1847–1996. International Journal for Parasitology 27: 1465–1480. PMID: <u>9467732</u>
- WHO (2011) Schistosomiasis: progress report 2001–2011 and strategic plan 2012–2020. Geneva: WHO.
- 54. WHO (2009) Elimination of schistosomiasis from low-transmission areas: report of a WHO informal consultation. Geneva: WHO.
- 55. Tambo E, Ai L, Zhou X, Chen JH, Hu W, et al. (2014) Surveillance-response systems: the key to elimination of tropical diseases. Infect Dis Poverty 3: 17. doi: 10.1186/2049-9957-3-17 PMID: 24971165
- Zhou XN, Bergquist R, Tanner M (2013) Elimination of tropical disease through surveillance and response. Infect Dis Poverty 2: 1. doi: <u>10.1186/2049-9957-2-1</u> PMID: <u>23849433</u>